Author's response to reviews

Title: Protective effect of Xuebijing injection on paraquat-induced pulmonary injury via down-regulating the expression of p38 MAPK

Authors:

Ming-wei Liu (lmw2004210@126.com)
Chuan-yun Qian (15198852569@139.com)

Version: 2
Date: 11 December 2013

Author's response to reviews: see over
Dear editor:

I am writing about MS: 1911903894115370. According to your comments, we made changes to the manuscript. Modifications as follows:

1. Title Page contain, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
2. Competing Interests was placed after the Conclusions/Abbreviations.
3. Authors’ Contributions was placed after the Competing Interests section.
4. Abbreviations was was placed after the Conclusions.
5. Funding was was placed after the Acknowledgments.

Thank you for your guidance,

Kind regards,

Dr Liu Ming-wei